Author’s response to reviews

Title: Frequency and determinants of thyroid autoimmunity in Ghanaian Type 2 diabetes patients: a case-control study

Authors:
OSEI SARFO-KANTANKA (osarfokantanka21@gmail.com)
STEPHEN SARFO (stephensarfo78@gmail.com)
EUNICE ANSAH (oparebeaansah@yahoo.com)
ERNEST YORKE (pavlovium@gmail.com)
JOSEPHINE AKPALU (janduah@yahoo.com)
BERNARD NKUM (bcnkum10@yahoo.co.uk)
BENJAMIN EGHAN (eghanben@gmail.com)

Version: 2 Date: 27 Dec 2016

Author’s response to reviews:

RESPONSE TO REVIEWERS

1. The manuscript has been revised to include all the sections requested under Declarations with consent for publication, availability of data and material section and Funding sections.

2. Consent for Publication was sought and has been described in the Declaration section, so the Supporting data is currently acceptable.